Author’s response to reviews

Title: Disclosure to genetic relatives without consent – Australian genetic professionals’ awareness of the health privacy law

Authors:

Natalia Meggiolaro (natalia.meggiolaro@gmail.com)
Kristine Barlow-Stewart (kristine.barlowstewart@sydney.edu.au)
Kate Dunlop (Kate.Dunlop@health.nsw.gov.au)
Ainsley Newson (ainsley.newson@sydney.edu.au)
Jane Fleming (jane.fleming@sydney.edu.au)

Version: 3 Date: 16 Jan 2020

Author’s response to reviews:

The edits requested have been completed.

1. References to additional files and tables have been addressed and edited.

2. The manuscript has been proof read and all tracking and other formatting has been removed.

Thank you

Kris Barlow-Stewart